**Author's response to reviews**

**Title:** Circulating CD36 and oxLDL levels are associated with cardiovascular risk factors in young subjects: a case-control study.

**Authors:**

Luz E Ramos-Arellano (luzelena_qbp@yahoo.com.mx)
José F Muñoz-Valle (biologiamolecular@hotmail.com)
Ulises De la Cruz-Mosso (ulises_cdm@hotmail.com)
Aralia B Salgado-Bernabé (araliux_27@hotmail.com)
Natividad Castro-Alarcón (natycastro2@hotmail.com)
Isela Parra-Rojas (iprojas@yahoo.com)

**Version:** 3

**Date:** 15 April 2014

**Author's response to reviews:** see over
April 14, 2014.

Dear Editor,

Thanks for your comments to improve our paper “Circulating CD36 and oxLDL levels are associated with cardiovascular risk factors in young subjects”, by the authors: Luz Elena Ramos-Arellano, José Francisco Muñoz-Valle, Ulises De la Cruz-Mosso, Aralia Berenice Salgado-Bernabé, Natividad Castro-Alarcón and Isela Parra-Rojas. Editor’s comments are answered below.

1. The title would be more appropriate if shortened to "Circulating CD36 and oxLDL levels are associated with cardiovascular risk factors in young subjects”.

   Answer: the title was modified.

2. Please include in the limitations the small number of subjects, especially obese as suggested by reviewer 2.

   Answer: the sentence was included in paragraph 2 of page 9.

3. Requesting for Copy-Edit.

   Answer: the article has been reviewed by a colleague who dominates the English language.

With best wishes,

Isela Parra Rojas, PhD.
Unidad Académica de Ciencias Químico Biológicas
Universidad Autónoma de Guerrero.